Author’s response to reviews

Title: Vitamin D concentrations and disease activity in Moroccan children with juvenile idiopathic arthritis

Authors:

Ilham Bouaddi (drbouaddilham@yahoo.fr)
Samira Rostom (samirostomsamira2003@yahoo.fr)
Dalal El badri (dalal_dr@yahoo.fr)
Asmae Hassani (hasmae00@hotmail.com)
Bouchra Chkirate (drbouaddilham@yahoo.fr)
Redoine Abouqal (drbouaddilham@yahoo.fr)
Bouchra Amine (amine_bouchra@yahoo.fr)
Najia hajjaj-hassouni (nhajjajhassouni@gmail.com)

Version: 4 Date: 7 March 2014

Author’s response to reviews: see over
Subject: SUBMISSION OF NEW MANUSCRIPT FOR EVALUATION

I am enclosing herewith a manuscript entitled “Vitamin D concentrations and disease activity in Moroccan children with juvenile idiopathic arthritis” submitted to “BMC musculoskeletal disorders” for possible evaluation.

With the submission of this manuscript i would like to undertake that the above mentioned manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere; and that my Institute’s El Ayachi hospital representative is fully aware of this submission.

In adults, many studies have shown an inverse association between disease activity and serum levels of 25(OH)D. However, data is limited regarding the association between disease activity and serum levels of 25(OH)D in children with juvenile idiopathic arthritis, so the aim of our study was to examine the association between serum levels of 25(OH)D and disease activity in Moroccan children with JIA.

Thank you for your time and consideration.

Sincerely,

Ilham Bouaddi